Author's response to reviews

Title: Occupational exposure and sinonasal cancer: a systematic review and meta-analysis

Authors:

Alessandra Binazzi (a.binazzi@inail.it)
Pierpaolo Ferrante (p.ferrante@inail.it)
Alessandro Marinaccio (a.marinaccio@inail.it)

Version: 7 Date: 12 January 2015

Author's response to reviews: see over
Answer to the Editor's Additional Request:

1. Please remove "Title:" and quotation marks from your title in our submission system and your title page.

2. Please include an acknowledgements section after your authors contributions. This should include funding details for the study and each author. If you received no funding, please state so in an acknowledgements section.

1) The word ‘Title’ and quotation marks have been removed from the submission system and in the title page.

2) An Acknowledgments section has been included to state that no funding source was received for this study and that no other person was involved, except for the authors.